\documentstyle[12pt,twoside]{article}
\input{macros-course}

\begin{document}

\handout{student-info}{Student Information Form}{February 1, 1998}

\begin{center}
{\bf Please return this sheet to us at the end of lecture today.}
\end{center}
\vskip 0.2in

We use information requested below to determine recitation section
assignments.  We will also provide an online class directory to help you
find classmates to collaborate with on problem sets.  If you do {\em not}
want to be listed in the online directory, please write ``NO LISTING'':

\medskip

\def\lineitem#1{
\hbox to \hsize
  {\hbox{#1} \hrulefill }
\medskip}

\lineitem{Last-name, First-name Middle}
\lineitem{Department and year}
\lineitem{Email address}
\lineitem{Phone number}

\bigskip

Each student must attend a two-hour recitation/tutorial on Friday.  To
schedule these, we ask you to {\bf cross out} those time slots in conflict
with your {\em nondiscretionary} commitments (eg, another course, team
practice) and {\bf circle} your preferred time slots; you may write
numbers beneath the circles to indicate order of preference.  Section
assignments will be distributed in lecture and posted on the course web
page on Thursday, February 5.

\vskip 0.2in
{\qquad 8--10 \qquad 9--11 \qquad 10--12 \qquad 11--1  \qquad 1--3 \qquad 2--4 \qquad 3--5 \qquad 4--6}

\vskip 0.7in

Please list all Course 18 and Course 6 Math/Theory subjects you have taken
or are currently taking.  Circle those you are currently taking.

\end{document}







